Author's response to reviews

**Title:** Cervicothoracic spinal cord and pontomedully injury secondary to high-voltage electrocution: a case report.

**Authors:**

Harpreet K. Johl (harpreetjohl@gmail.com)  
Adel Olshansky (adel.olshansky@gmail.com)  
Said R. Beydoun (sbeydoun@usc.edu)  
Richard A. Rison (rison@usc.edu)

**Version:** 6 **Date:** 19 July 2012

**Author's response to reviews:** see over
July 19th, 2012

Dear JMCR Editorial Team,

We thank the editorial staff for their comments in further reviewing our manuscript. All changes have been made as suggested:

1. Please make sure that author details are only confined in the first page of the manuscript—this has now been done.
2. Please replace the term “gentleman” with “male”—this has now been done.
3. Please remove the “watermarked” text running along the middle of the manuscript pages—the manuscript was formatted again and we do not see any “watermarked” text.

Sincerely yours,

Richard A. Rison, M.D.